Author's response to reviews

Title: Cronkhite-Canada syndrome associated with rib fractures: a case report.

Authors:

Yuan Bosi (cat409@126.com)
Jin Xinxin (jinxx1973@sina.com)
Zhu Renmin (zhurm@126.com)
Zhang Xiaohua (jszhxh@sina.com)
Liu Jiong (Liujiong123@sina.com)
Wan Haijun (xsx1980ren@sohu.com)
Lu Heng (land9956@yahoo.com.cn)
Shen Yunzhu (yunzhushen@gmail.com)
Wang Fangyu (wangfy65@gmail.com)

Version: 3 Date: 22 September 2010

Author's response to reviews: see over
Dear Dr. Tim,

Thank you for returning the referee’s comment on our paper by Yuan et al entitled ‘Cronkhite-Canada syndrome associated with rib fractures: a case report’.

According to his/her comment, we have carefully checked grammatically incorrect sentences and revised them. We are grateful to this referee for his/her thorough checking on our manuscript.

Finally, we would like to thank two referees for their highly constructive suggestions to improve this manuscript in the acknowledgement section if they permit. Could you kindly provide their full name for me as they are not shown very clearly on referee’ comment?

We look forward to hearing from you and hope that these modifications will help our manuscript to be accepted by BMC Gastroenterology.

Best regards,

Yuan bosi on behalf of Dr. Wang Fangyu